Author's response to reviews

Title: Effects of Scleral Collagen Cross-linking using Genipin on the Process of Form-deprivation Myopia in the Guinea Pig: a randomized controlled experimental study

Authors:

Mengmeng Wang (wangmengmg@163.com)
Christine Carole C. Corpuz (1417325260@qq.com)

Version: 3 Date: 11 April 2015

Author's response to reviews: see over
April, 11th, 2015

Dear editor,
Thank you for your review of our article "Effects of Scleral Collagen Cross-linking using Genipin on the Process of Form-deprivation Myopia in the Guinea Pig: a randomized controlled experimental study" (MS: 8690345491673956).

Accordingly, the following changes has been made in the new reviewed manuscript.

1. Competing Interest
   The "Competing Interest" section has been added after the Conclusions in the new manuscript.
   "The authors declare that they have no competing interests."

2. Author's Contribution
   The "Author's Contribution" section has been added after the Competing Interests section in the new manuscript.
   "Mengmeng Wang made substantial contributions to conception and design of the study, acquisition and analysis of data. Mengmeng Wang carried out the FD and genipin CXL procedures, and biological measurements. Christine Carole C. Corpuz participated in the light and electron microscopy, and helped to draft the manuscript. All authors read and approved the final manuscript."

3. Acknowledgments
   The "Acknowledgments" section has been added after the Author's Contribution section in the new manuscript.
   "There is no acknowledgment to be declared in this study."

4. Requesting name of ethics committee
   The name of ethics committee has been added in the first paragraph (line 64, Page 5) of Materials and Methods.
   "The study adheres to the Association for Research in Vision and Ophthalmology statement for the use of animals in ophthalmology and vision research and was approved by the Ethical Committee of Peking University."

If any question, please feel free to contact us.

Best wishes,

Mengmeng Wang